Author’s response to reviews

Title: The Association between Oxidative Stress and Endothelial Dysfunction in Early Childhood Patients with Kawasaki Disease

Authors:
takamichi ishikawa (ishikawa@hama-med.ac.jp)
Keigo Seki (tbb0726@yahoo.co.jp)

Version: 6 Date: 14 Jan 2018

Author’s response to reviews:

January 14, 2018

Francesca Dickens
Editor, BMC Cardiovascular Disorders

RE: Manuscript for BMC Cardiovascular Disorders #BCAR-D-17-00368R5 by T. Ishikawa et al.

Dear Dr. Dickens,

Thank you very much for your email. We have completely revised our manuscript according to the suggestions.

Thank you for your consideration of the revised version of our manuscript.

Respectfully yours,

Takamichi Ishikawa, MD, PhD.
Department of Pediatrics, Hamamatsu University School of Medicine, 1-20-1 Handayama, Higashi-ku, Hamamatsu 431-3192, Japan

Telephone; +81-053-435-2312, Fax; +81-053-435-2311,

E-mail; ishikawa@hama-med.ac.jp